ANSWERING REVIEWERS

Name of journal: World Journal of Methodology

Manuscript NO: 76541

Title: Syndemic aspects between the COVID-19 pandemic and social inequality: a minireview

Reviewer’s code: 03117998

Position: Anonymous

Dear reviewer,

We are glad for the great evaluation and suggestions. Your suggestions were very useful in the improvement of the scientific quality of this article. Aiming to attend to your suggestions, the following changes were made:

1- The search methodology for the manuscript elaboration was added;
2- We made a table describing the main findings of the theme;

Best regards,

Fabrício Freire de Melo

Professor, PhD

Reviewer’s code: 04159984

Position: Peer Reviewer

Dear reviewer,
Thank you very much for your comment. Aiming to improve the quality of our manuscript and attend to your requirement all alterations and suggestions are listed below:

1. We made a new part about the inequalities in vaccine access;
2. The effects of the pandemic on the indigenous populations of the Brazil and world were added to the article;
3. A new text was written to address the relation between COVID-19 and the population on the war stress;

Best regards,

Fabrício Freire de Melo

Professor, PhD

**Reviewer’s code:**

**Position:** Editor-in-Chief

Dear reviewer,

We are glad about your suggestions. We really appreciate them and the alterations are listed the alterations.

1. We produce a table including the main impacts of the pandemic on the different populations;

Best regards,

Fabrício Freire de Melo

Professor, PhD

**Reviewer’s code:**
Dear reviewer,

Thank you for your comments. We really appreciate them. Below are the alterations that you suggested.

1. We rewrite the abstract section;
2. The methodology for the construction of this manuscript was added;

Best regards,

Fabrício Freire de Melo

Professor, PhD